Author's response to reviews

Title: Gangrenous appendicitis presenting as acute abdominal pain in a patient on automated peritonal dialysis: a case report

Authors:

Robert Ekart (robert.ekart2@guest.arnes.si)
Matjaz Horvat (maco.horvat@gmail.com)
Miran Kozelj (miran.kozelj@ukc-mb.si)
Breda Pecovnik Balon (breda.balon@ukc-mb.si)
Sebastjan Bevc (sebastjanbevc@yahoo.com)
Radovan Hojs (radovan.hojs@ukc-mb.si)

Version: 5 Date: 4 August 2012

Author's response to reviews: see over
Re-Submission date: 04th August 2012

Manuscript Number: 1395263437683120

Journal of Medical Case Reports

Dear Editor,

According to your request we have changed and replaced the title of our manuscript with:

"Gangrenous appendicitis presenting as acute abdominal pain in a patient on automated peritoneal dialysis: a case report"

We hope, that this change now conform all your requirements.

With kind regards,
on behalf of the co-authors

Robert Ekart